Author's response to reviews

Title: An Analysis of Rural Residents’ Perceptions of the New Cooperative Medical System in Liaoning Province, China

Authors:

Xi Long Pan (xilongpan@126.com)
Ying Zhang (zhangying2006100@126.com)
Qian Qian Zhao (qianqian_zhao1984@126.com)
Ju Huang (lihuihuang-1@163.com)
Li Xu (xuli_530@126.com)

Version: 7 Date: 26 July 2009

Author's response to reviews:

The approval of an appropriate ethics committee has been added to the Methods section of the manuscript, including the name of the body which gave approval, with a reference number where appropriate.